Massachusetts Institute of Technology Kinetic Processes in Materials

3.21 Spring 2002 Samuel M. Allen and W. Craig Carter

Department of Materials Science and Engineering Massachusetts Institute of Technology 77 Massachusetts Ave. Cambridge, MA 02139

Problem Set 8: Due Wed. April 24, 2002, emailed to ccarter@mit.edu

Exercise 8.1

Write a short poem of any style and any language of your choice about nucleation and growth.

If you wish to have your poem published^1 then email your poem to ccarter@mit.edu and state that you allow publication of your poem.

1please see http://pruffle.mit.edu/c̃carter/3.21/poetry/